\def\url#1{\texttt{{\small #1}}}
\begin{thebibliography}{10}

\bibitem{Alae94:Design}
C.~Alaettino\u{g}lu, A.U. Shankar, K.~Dussa-Zeiger, and I.~Matta.
\newblock Design and implementation of {MaRS}: A routing testbed.
\newblock {\em Internetworking: Research and Experience}, 5:17--41, 1994.

\bibitem{Bajaj99a}
Sandeep Bajaj, Lee Breslau, Deborah Estrin, Kevin Fall, Sally Floyd, Padma
  Haldar, Mark Handley, Ahmed Helmy, John Heidemann, Polly Huang, Satish Kumar,
  Steven McCanne, Reza Rejaie, Puneet Sharma, Kannan Varadhan, Ya~Xu, Haobo Yu,
  and Daniel Zappala.
\newblock Improving simulation for network research.
\newblock Technical Report 99-702b, University of Southern California, March
  1999.
\newblock (revised September 1999).

\bibitem{Barf98:WebWorkload}
Paul Barford and Mark Crovella.
\newblock Generating representative web workloads for network and server
  peformance evaluation.
\newblock In {\em Proceedings of the~ACM SIGMETRICS}, pages 151--160, June
  1998.

\bibitem{Brak94:TCP}
L.S. Brakmo, S.~O'Malley, and L.L. Peterson.
\newblock {TCP} vegas: New techniques for congestion detection and avoidance.
\newblock In {\em Proceedings of the~ACM SIGCOMM}, pages 24--35, October 1994.

\bibitem{Brak94a:TCP}
L.S. Brakmo, S.~O'Malley, and L.L. Peterson.
\newblock {TCP} vegas: New techniques for congestion detection and avoidance.
\newblock Technical Report TR 94 04, Department of Computer Science, The
  University of Arizona, Tucson, AZ 85721, February 1994.

\bibitem{Brow88:Calendar}
R.~Brown.
\newblock Calendar queues: A fast {O(1)} priority queue implementation for the
  simulation event set problem.
\newblock {\em Communications of the ACM}, 31(10):1220--1227, October 1988.

\bibitem{Cao97:CacheConsistency}
Pei Cao and Chengjie Liu.
\newblock Maintaining strong cache consistency in the {World-Wide Web}.
\newblock In {\em Proceedings of the~IEEE ICDCS}, pages 12--21, May 1997.

\bibitem{ChLiAb02a}
N.~Christin, J.~Liebeherr, and T.~Abdelzaher.
\newblock A quantitative assured forwarding service.
\newblock In {\em Proceedings of {IEEE INFOCOM 2002}}, volume~2, pages
  864--873, New York, NY, June 2002.

\bibitem{Deer94a:Architecture}
S.~Deering, D.~Estrin, D.~Farinacci, V.~Jacobson, Ching-{G}ung Liu, and L.~Wei.
\newblock An architecture for wise-area multicast routing.
\newblock Technical Report USC-SC-94-565, Computer Science Department,
  University of Southern California, Los Angeles, CA 90089., 1994.

\bibitem{Feldmann99a}
Anja Feldmann, Anna~C. Gilbert, Polly Huang, and Walter Willinger.
\newblock Dynamics of {IP} traffic: A study of the role of variability and the
  impact of control.
\newblock pages 301--313, Cambridge, MA, USA, August 1999.

\bibitem{Floy95:Reliable}
S.~Floyd, V.~Jacobson, C.~Liu, S.~McCanne, and L.~Zhang.
\newblock A reliable multicast framework for light-weight sessions and
  application level framing.
\newblock In {\em Proceedings of the~ACM SIGCOMM}, pages 342--356, August 1995.

\bibitem{Friis46}
H.~T. Friis.
\newblock A note on a simple transmission formula.
\newblock {\em Proc. IRE}, 34, 1946.

\bibitem{Hethcote00inf}
H.~W. Hethcote.
\newblock The mathematics of infectious diseases.
\newblock {\em SIAM Review}, 42(4):599--653, October 2000.

\bibitem{Heyb89:Netsim}
A.~Heybey.
\newblock {\em Netsim Manual}.
\newblock MIT, 1989.

\bibitem{Jain91:Art}
R.~Jain.
\newblock {\em The Art of Computer Systems Performance Analysis}.
\newblock John Wiley and Sons, Inc., 1996.

\bibitem{lecuyer99}
Pierre L'Ecuyer.
\newblock Good parameters and implementations for combined multiple recursive
  random number generators.
\newblock {\em Operations Research}, 47(1):159--164, 1999.

\bibitem{lecuyer-wsc}
Pierre L'Ecuyer.
\newblock Software for uniform random number generation: Distinguishing the
  good and the bad.
\newblock In {\em Proceedings of the 2001 Winter Simulation Conference}, pages
  95--105, December 2001.

\bibitem{lecuyer01}
Pierre L'Ecuyer, Richard Simard, E.~Jack Chen, and W.~David Kelton.
\newblock An object-oriented random number package with many long streams and
  substreams.
\newblock {\em Operations Research}, 2001.

\bibitem{legout_sigmetrics2000}
A.~Legout and E.W. Biersack.
\newblock {PLM}: Fast convergence for cumulative layered multicast transmission
  schemes.
\newblock In {\em Proceedings of the~ACM SIGMETRICS}, Santa Clara, CA, U.S.A.,
  June 2000.

\bibitem{LiCh01}
J.~Liebeherr and N.~Christin.
\newblock {JoBS}: Joint buffer management and scheduling for differentiated
  services.
\newblock In {\em Proceedings of {IWQoS 2001}}, pages 404--418, Karlsruhe,
  Germany, June 2001.

\bibitem{LiCh02}
J.~Liebeherr and N.~Christin.
\newblock Rate allocation and buffer management for differentiated services.
\newblock {\em Computer Networks}, 40(1):89--110, September 2002.

\bibitem{Math96:Forward}
M.~Mathis and J.~Mahdavi.
\newblock Forward acknowledgement: Refining {TCP} congestion control.
\newblock In {\em Proceedings of the~ACM SIGCOMM}, August 1996.

\bibitem{rfc2018}
M.~Mathis, J.~Mahdavi, S.~Floyd, and A.~Romanov.
\newblock {\em {TCP} Selective Acknowledgement Options}, {RFC 2018} edition,
  1996.

\bibitem{ns}
S.~McCanne and S.~Floyd.
\newblock ns---{N}etwork {S}imulator.
\newblock \url{http://www-mash.cs.berkeley.edu/ns/}.

\bibitem{BPF93}
S.~McCanne and V.~Jacobson.
\newblock The bsd packet filter: A new architecture for user-level packet
  capture.
\newblock pages 259--269, January 1993.

\bibitem{Ousterhout98a}
John Ousterhout.
\newblock Scripting: Higher-level programming for the 21st century.
\newblock {\em {IEEE} Computer}, 31(3):23--30, March 1998.

\bibitem{Park88:Random}
S.K. Park and R.W. Miller.
\newblock Random number generation: Good ones are hard to find.
\newblock {\em Communications of the ACM}, 31(10):1192--1201, October 1988.

\bibitem{Diffserv}
Peter Pieda, Jeremy Ethridge, Mandeep Baines, and Farhan Shallwani.
\newblock {\em A Network Simulator, Differentiated Services Implementation}.
\newblock Open IP, Nortel Networks, 2000.

\bibitem{Rappaport96}
T.~S. Rappaport.
\newblock {\em Wireless communications, principles and practice}.
\newblock Prentice Hall, 1996.

\bibitem{rfc1075}
D.~Waitzman, C.~Partridge, and S.E. Deering.
\newblock {\em {Distance Vector Multicast Routing Protocol}}, {RFC 1075}
  edition, 1988.

\bibitem{Katabi02}
Dina Katabi, Mark Handley, and Chalrie Rohrs.
\newblock Congestion Control for High Bandwidth-Delay Product Networks
\newblock ACM Sigcomm 2002

\end{thebibliography}

%
% personal commentary:
%        DRAFT DRAFT DRAFT
%        - KFALL
% RNG section reviewed and revised 11-Sep-98, johnh.
%
% RNG section revised 28-Aug-02 by Tim Buchheim to include material
% from Michelle Weigle which describes the new implementation of RNG

\chapter{Mathematical Support}
\label{chap:math}

The simulator includes a small collection of mathematical
functions used to implement random variate generation and integration.
This area of the simulator is currently undergoing some
changes.

The procedures and functions described in this chapter can be found in
\nsf{tools/rng.\{cc, h\}},
\nsf{tools/random.\{cc, h\}},
\nsf{tools/ranvar.\{cc, h\}},
\nsf{tools/pareto.\{cc, h\}},
\nsf{tools/expoo.\{cc, h\}},
\nsf{tools/integrator.\{cc, h\}}, and
\nsf{tcl/lib/ns-random.tcl}.


\section{Random Number Generation}
\label{sec:random}

The RNG class contains an implementation of the combined multiple
recursive generator MRG32k3a proposed by L'Ecuyer
\cite{lecuyer99}. The C++ code was adapted from \cite{lecuyer01}.
This replaces the previous implementation of \code{RNG}, which used
the minimal standard multiplicative linear congruential generator of
Park and Miller~\cite{Park88:Random}.  The newer (MRG32k3a) \code{RNG} is
used in ns versions 2.1b9 and later.

The MRG32k3a generator provides $1.8$x$10^{19}$ independent
streams of random numbers, each of which consists of
$2.3$x$10^{15}$ substreams. Each substream has a period
(\emph{i.e.}, the number of random numbers before overlap) of
$7.6$x$10^{22}$. The period of the entire generator is
$3.1$x$10^{57}$. Figure \ref{streams} provides a graphical idea of
how the streams and substreams fit together.
\begin{figure}[h]
\centering
\includegraphics[angle=270,width=6 in]{figures/rng-streams.eps}
\caption{Overall arrangement of streams and substreams.
\cite{lecuyer01}} \label{streams}
\end{figure}

A default RNG ({\tt defaultRNG}), created at simulator initialization
time, is provided. If multiple random variables are used in a
simulation, each random variable should use a separate RNG object.
When a new RNG object is created, it is automatically seeded to
the beginning of the next independent stream of random numbers.
Used in this manner, the implementation allows for a maximum of
$1.8$x$10^{19}$ random variables.

Often, multiple independent replications of a simulation are
needed (\emph{i.e.}, to perform statistical analysis given
multiple runs with fixed parameters).  For each replication, a
different substream should be used to ensure that the random
number streams are independent. (This process is given as an OTcl
example later.) This implementation allows for a maximum of
$2.3$x$10^{15}$ independent replications. Each random variable in
a single replication can produce up to $7.6$x$10^{22}$ random
numbers before overlapping.

{\bf Note:} Only the most common functions are described here. For
more information, see \cite{lecuyer01} and the source code found
in {\tt tools/rng.h} and {\tt tools/rng.cc}.  For a comparison of this RNG to
the more common LCG16807 RNG (and why LCG16807 is not a good RNG), 
see \cite{lecuyer-wsc}.
\subsection{Seeding The RNG}
Due to the nature of the RNG and its implementation, it is not
necessary to set a seed (the default is 12345).  If you wish to
change the seed, functions are available. You should only set the
seed of the default RNG.  Any other RNGs you create are
automatically seeded such that they produce independent streams.
The range of valid seeds is 1 to {\tt MAXINT}.

To get non-deterministic behavior, set the seed of the default RNG
to 0.  This will set the seed based on the current time of day and
a counter.  {\bf This method should not be used to set seeds for
independent replications.}  There is no guarantee that the streams
produced by two random seeds will not overlap.  The only way to
guarantee that two streams do not overlap is to use the substream
capability provided by the RNG implementation.

\subsubsection{Example}
\begin{verbatim}
 # Usage: ns rng-test.tcl [replication number]

 if {$argc > 1} {
    puts "Usage: ns rng-test.tcl \[replication number\]"
    exit
 }
 set run 1
 if {$argc == 1} {
    set run [lindex $argv 0]
 }
 if {$run < 1} {
    set run 1
 }

 # seed the default RNG
 global defaultRNG
 $defaultRNG seed 9999

 # create the RNGs and set them to the correct substream
 set arrivalRNG [new RNG]
 set sizeRNG [new RNG]
 for {set j 1} {$j < $run} {incr j} {
    $arrivalRNG next-substream
    $sizeRNG next-substream
 }

 # arrival_ is a exponential random variable describing the time
 # between consecutive packet arrivals
 set arrival_ [new RandomVariable/Exponential]
 $arrival_ set avg_ 5
 $arrival_ use-rng $arrivalRNG

 # size_ is a uniform random variable describing packet sizes
 set size_ [new RandomVariable/Uniform]
 $size_ set min_ 100
 $size_ set max_ 5000
 $size_ use-rng $sizeRNG

 # print the first 5 arrival times and sizes
 for {set j 0} {$j < 5} {incr j} {
    puts [format "%-8.3f  %-4d" [$arrival_ value] \
            [expr round([$size_ value])]]
 }
\end{verbatim}

\subsubsection{Output}

\begin{verbatim}
 % ns rng-test.tcl 1
 6.358     4783
 5.828     1732
 1.469     2188
 0.732     3076
 4.002     626

 % ns rng-test.tcl 5
 0.691     1187
 0.204     4924
 8.849     857
 2.111     4505
 3.200     1143
\end{verbatim}

\subsection{OTcl Support}

\subsubsection{Commands}

The following commands on the RNG class can be accessed from OTcl
and are found in {\tt tools/rng.cc}:
\begin{description}
    \item[{\tt seed $n$}] -- seed the RNG to $n$, if $n == 0$, the seed
    is set according to the current time and a counter
    \item[{\tt next-random}] -- return the next random number
    \item[{\tt seed}] -- return the current value of the seed
    \item[{\tt next-substream}] -- advance to the next substream
    \item[{\tt reset-start-substream}] -- reset the stream to the beginning
    of the current substream
    \item[{\tt normal $avg$ $std$}] -- return a number sampled from a normal
    distribution with the given average and standard deviation
    \item[{\tt lognormal $avg$ $std$}] -- return a number sampled from a
      lognormal distribution with the given average and standard deviation
\end{description}

The following commands on the RNG class can be accessed from OTcl
and are found in {\tt tcl/lib/ns-random.tcl}:
\begin{description}
    \item[{\tt exponential $mu$}] -- return a number sampled from an
      exponential distribution with mean $mu$ 
    \item[{\tt uniform $min$ $max$}] -- return an integer sampled from a
    uniform distribution on [$min$, $max$]
    \item[{\tt integer $k$}] -- return an integer sampled from a uniform
    distribution on [0, $k$-1]
\end{description}

\subsubsection{Example}

\begin{verbatim}
 # Usage: ns rng-test2.tcl [replication number]

 if {$argc > 1} {
    puts "Usage: ns rng-test2.tcl \[replication number\]"
    exit
 }
 set run 1
 if {$argc == 1} {
    set run [lindex $argv 0]
 }
 if {$run < 1} {
    set run 1
 }

 # the default RNG is seeded with 12345

 # create the RNGs and set them to the correct substream
 set arrivalRNG [new RNG]
 set sizeRNG [new RNG]
 for {set j 1} {$j < $run} {incr j} {
    $arrivalRNG next-substream
    $sizeRNG next-substream
 }

 # print the first 5 arrival times and sizes
 for {set j 0} {$j < 5} {incr j} {
    puts [format "%-8.3f  %-4d" [$arrivalRNG lognormal 5 0.1] \
            [expr round([$sizeRNG normal 5000 100])]]
 }
\end{verbatim}

\subsubsection{Output}

\begin{verbatim}
 % ns rng-test2.tcl 1
 142.776   5038
 174.365   5024
 147.160   4984
 169.693   4981
 187.972   4982

 % ns rng-test2.tcl 5
 160.993   4907
 119.895   4956
 149.468   5131
 137.678   4985
 158.936   4871
\end{verbatim}

\subsection{C++ Support}

\subsubsection{Member Functions}
The random number generator is implemented by the RNG class and is
defined in {\tt tools/rng.h}.

{\bf Note:} The Random class in {\tt tools/random.h} is an older
interface to the standard random number stream.

Member functions provide the following operations:

\begin{description}
    \item[{\tt void set\_seed (long seed)}] -- set the seed of the RNG, if
    $seed == 0$, the seed is set according to the current time and a
    counter
    \item[{\tt long seed (void)}] -- return the current seed
    \item[{\tt long next (void)}] -- return the next random number as an
    integer on [0, {\tt MAXINT}]
    \item[{\tt double next\_double (void)}] -- return the next random number
    on [0, 1]
    \item[{\tt void reset\_start\_substream (void)}] -- reset the stream
    to the beginning of the current substream
    \item[{\tt void reset\_next\_substream (void)}] -- advance to the next
    substream
    \item[{\tt int uniform (int k)}] -- return an integer sampled from a
      uniform distribution on [0, k-1]
    \item[{\tt double uniform (double r)}] -- return a number sampled from a
      uniform distribution on [0, r]
    \item[{\tt double uniform (double a, double b)}] -- return a number
      sampled from a uniform distribution on [a, b]
    \item[{\tt double exponential (void)}] -- return a number sampled from an
    exponential distribution with mean 1.0
    \item[{\tt double exponential (double k)}] -- return a number sampled from
      an exponential distribution with mean k 
    \item[{\tt double normal (double avg, double std)}] -- return a number
      sampled from a normal distribution with the given average and standard 
    deviation
    \item[{\tt double lognormal (double avg, double std)}] -- return a number
      sampled from a lognormal distribution with the given average and
      standard deviation
\end{description}

\subsubsection{Example}

\begin{verbatim}
 /* create new RNGs */
 RNG arrival (23456);
 RNG size;

 /* set the RNGs to the appropriate substream */
 for (int i = 1; i < 3; i++) {
   arrival.reset_next_substream();
   size.reset_next_substream();
 }

 /* print the first 5 arrival times and sizes */
 for (int j = 0; j < 5; j++) {
   printf ("%-8.3f  %-4d\n", arrival.lognormal(5, 0.1),
             int(size.normal(500, 10)));
 }
\end{verbatim}

\subsubsection{Output}
\begin{verbatim}
 161.826   506
 160.591   503
 157.145   509
 137.715   507
 118.573   496
\end{verbatim}





\section{Random Variables}
\label{sec:ranvar}

The \clsref{RandomVariable}{../ns-2/ranvar.h}
provides a thin layer of functionality on top
of the base random number generator and the default random number stream.
It is defined in \nsf{ranvar.h}:

\begin{program}
  class RandomVariable : public TclObject \{
  public:
        virtual double value() = 0;
        int command(int argc, const char*const* argv);
        RandomVariable();
  protected:
        RNG* rng_;
  \};
\end{program}

Classes derived from this abstract class implement specific
distributions.  Each distribution is parameterized with the values of
appropriate parameters.  The value method is used to return a value
from the distribution.  

The currently defined distributions, and their associated parameters are:

\begin{tabular}{rl}
\clsref{UniformRandomVariable}{tools/ranvar.h} & \code{min_}, \code{max_} \\
\clsref{ExponentialRandomVariable}{tools/ranvar.h} & \code{avg_} \\
\clsref{ParetoRandomVariable}{tools/ranvar.h} & \code{avg_}, \code{shape_}\\
\clsref{ParetoIIRandomVariable}{tools/ranvar.h} & \code{avg_}, \code{shape_}\\
\clsref{ConstantRandomVariable}{tools/ranvar.h} & \code{val_}\\
\clsref{HyperExponentialRandomVariable}{tools/ranvar.h} & \code{avg_}, \code{cov_}\\
\clsref{NormalRandomVariable}{tools/ranvar.h} & \code{avg_}, \code{std_}\\
\clsref{LogNormalRandomVariable}{tools/ranvar.h} & \code{avg_}, \code{std_}\\
\end{tabular}

The RandomVariable class is available in OTcl.  For instance, to
create a random variable that generates number uniformly on [10, 20]:
\begin{program}
        set u [new RandomVariable/Uniform]
        $u set min_ 10
        $u set max_ 20
        $u value
\end{program}
By default, RandomVariable objects use the default random number
generator described in the previous section.  The use-rng method can
be used to associate a RandomVariable with a non-default RNG:
\begin{program}
        set rng [new RNG]
        $rng seed 0

        set e [new RandomVariable/Exponential]
        $e use-rng $rng
\end{program}


\section{Integrals}
\label{sec:integral}

The  \clsref{Integrator}{../ns-2/integrator.h}
supports the approximation of (continuous) integration by (discrete)
sums; it is defined in \nsf{integrator.h} as
\begin{program}
{\rm From integrator.h:}
        class Integrator : public TclObject \{
        public:
                Integrator();
                void set(double x, double y);
                void newPoint(double x, double y);
                int command(int argc, const char*const* argv);
        protected:
                double lastx_;
                double lasty_;
                double sum_;
        \};

{\rm From integrator.cc:}
        Integrator::Integrator() : lastx_(0.), lasty_(0.), sum_(0.)
        \{
                bind("lastx_", &lastx_);
                bind("lasty_", &lasty_);
                bind("sum_", &sum_);
        \}

        void Integrator::set(double x, double y)
        \{
                lastx_ = x;
                lasty_ = y;
                sum_ = 0.;
        \}

        void Integrator::newPoint(double x, double y)
        \{
                sum_ += (x - lastx_) * lasty_;
                lastx_ = x;
                lasty_ = y;
        \}

        int Integrator::command(int argc, const char*const* argv)
        \{
                if (argc == 4) \{
                        if (strcmp(argv[1], "newpoint") == 0) \{
                                double x = atof(argv[2]);
                                double y = atof(argv[3]);
                                newPoint(x, y);
                                return (TCL_OK);
                        \}
                \}
                return (TclObject::command(argc, argv));
        \}
\end{program}
This class provides a base class used by other classes such
as \code{QueueMonitor} that keep running sums.
Each new element of the running sum is added by
the \fcn[x, y]{newPoint} function.
After the $k$th execution of \code{newPoint}, the running sum
is equal to $\sum_{i=1}^{k}y_{i-1}(x_i - x_{i-1})$ where
$x_0 = y_0 = 0$ unless \code{lastx\_}, \code{lasty\_}, or \code{sum\_}
are reset via OTcl.
Note that a new point in the sum can be added either by the
C++ member \code{newPoint} or the OTcl member \code{newpoint}.
The use of integrals to compute certain types of averages
(e.g. mean queue lengths) is given in (pp. 429--430, \cite{Jain91:Art}).

\section{\code{ns-random}}

{\bf \code{ns-random} is an obsolete way to generate random numbers. 
This information is provided only for backward compatibility.}

\code{ns-random} is implemented in \nsf{misc.\{cc,h\}}. 
When called with no argument, it generates a random number with
uniform distribution between 0 and \code{MAXINT}.
When an integer argument is provided, it seeds the random generater
with the given number.
A special case is when \code{ns-random 0} is called, it randomly seeds
the generator based on current time.
This feature is useful to produce non-deterministic results across
runs.

\section{Some mathematical-support related objects}
\label{sec:mathobjects}

\textsc{Integrator objects}
Integrator Objects support the approximate computation of continuous
integrals using discrete sums. The running sum(integral) is computed as:
\code{sum_ += [lasty_ * (x lastx_)]} where (x, y) is the last element
entered and
(lastx\_, lasty\_) was the element previous to that added to the sum.
lastx\_ and lasty\_ are updated as new elements are added. The first
sample point defaults to (0,0) that can be changed by changing the values
of (lastx\_,lasty\_). 
\code{$integrator newpoint <x> <y>}\\ %$
Add the point (x,y) to the sum. Note that it does not make sense for x to
be less than lastx\_. 

There are no configuration parameters specific to this object. \\

State Variables are:
\begin{description}
\item[lastx\_]
x-coordinate of the last sample point. 

\item[lasty\_]
y-coordinate of the last sample point. 

\item[sum\_] Running sum (i.e. the integral) of the sample points. 
\end{description}


\textsc{Samples Object}
Samples Objects support the computation of mean and variance statistics
for a given sample. 

\code{$samples mean}\\
Returns mean of the sample. 

\code{$samples variance}\\
Returns variance of the sample. 

\code{$samples cnt}\\
Returns a count of the sample points considered. 

\code{$samples reset }\\
Reset the Samples object to monitor a fresh set of samples. 

There are no configuration parameters or state variables specific to this
object. 



\section{Commands at a glance}
\label{sec:mathcommand}

Following is a list of mathematical support related commands commonly used
in simulation scripts:
\begin{flushleft}
\code{set rng [new RNG]}\\
This creates a new random number generator.


\code{$rng seed <0 or n>}\\
This command seeds the RNG. If 0 is specified, the RNG is seeded
heuristically. Otherwise the RNG is seeded with the value <n>.


\code{$rng next-random}\\
This returns the next random number from RNG.


\code{$rng uniform <a> <b>}\\
This returns a number uniformly distributed on <a> and <b>.


\code{$rng integer <k>}\\
This returns an integer uniformly distributed on 0 and k-1.


\code{$rng exponential}\\
This returns a number that has exponential distribution with average 1.


\code{set rv [new Randomvariable/<type of random-variable>]}\\
This creates an instance of a random variable object that generates random 
variables with specific distribution. The different types of random 
variables derived from the base class are:\\
RandomVariable/Uniform, RandomVariable/Exponential, RandomVariable/Pareto,
RandomVariable/Constant, RandomVariable/HyperExponential.
Each of these distribution types are parameterized with values of
appropriate parameters. For details see section \ref{sec:ranvar} of this
chapter.


\code{$rv use-rng <rng>}\\
This method is used to associated a random variable object with a
non-default RNG. Otherwise by default, the random variable object is
associated with the default random number generator.


\end{flushleft}




% LocalWords:  ranvar pareto expoo tcl lib ns cc integrator RNG rng TclObject
% LocalWords:  enum RNGSources PREDEF int MAXINT rX edv prob edp setbit iph ECN
% LocalWords:  ip ecn eed OURCE OTcl RandomVariable argc const argv min avg val
% LocalWords:  UniformRandomVariable ExponentialRandomVariable cov newPoint pp
% LocalWords:  ParetoRandomVariable ConstantRandomVariable lastx lasty strcmp
% LocalWords:  HyperExponentialRandomVariable newpoint atof QueueMonitor
